Author's response to reviews

Title: Pelvic actinomycosis presenting with a large abscess and bowel stenosis with marked response to conservative treatment: a case report

Authors:

Hiroaki Nozawa (hiroanozawa-gi@umin.ac.jp)
Yoshinao Yamada (yoshinao.yamada.cy@hitachi.com)
Yasuhiko Muto (yasuhiko.muto.ve@hitachi.com)
Shirane Arita (shirane.arita hv@hitachi.com)
Kohzo Aisaka (kozo.aisaka tt@hitachi.com)

Version: 4 Date: 22 October 2007

Author's response to reviews:

Dear editors and reviewers,

I would appreciate very much if you could consider our manuscript entitled "Pelvic actinomycosis presenting with a large abscess and bowel stenosis with marked response to conservative treatment: a case report" for publication in "Journal of Medical Case Reports".

In this case report, MRI and a cervial smear are important for deciding the treatment strategy as a reviewer pointed out. The effects of conservative treatment were striking.

According to a reviewer's suggestion, the manuscript was critically checked by a native English speaker and revised. The title has also been changed so that the two clinically important findings, large abscess and bowel stenosis, can be highlighted. Now we believe that our revised manuscript is acceptable for the journal.

Thank you again for your kind consideration.

With my best regards,

Yours sincerely,

Hiroaki Nozawa, M.D., Ph.D.